Author's response to reviews

Title: Pre-referral rectal artesunate in severe malaria: a flawed trial

Authors:

Karim F Hirji (kfhirji@aol.com)
Zulfiqarali G Premji (zulpremji688@gmail.com)

Version: 7 Date: 29 March 2011

Author's response to reviews:

Dear Editorial Production Team:

A corrected version of the manuscript is provided. The two major revisions have been made. Please note the resultant change in the first paragraph of the abstract. Also, a few typographical errors were corrected.

Thank you for your kind assistance.

Karim F Hirji